Math 144 Fall 2006 R. Schultz

UPDATED GENERAL INFORMATION — NOVEMBER 20, 2006 Here is the sixth homework assignment, which is due in class on Monday, December 4, 2006. All references (including section numbers) are to the file math144exercises6-7.pdf.

Section VI.1: 3a, 4b Section VI.2: 1, 2 Section VI.3: 1, 2 Section VI.4: 1, 2, 4 Final ungraded assignment There will be no homework for Unit VII to be turned in for grading, but it is strongly recommended that the following assinged problems be worked out in detail just as if they would be graded.

Section VII.1: 1, 2 Section VII.2: 1–5 Section VII.3: 2 Section VII.4: 3